Author’s response to reviews

Title: Mechanism of tanshinones and phenolic acids from Danshen in the Treatment of Coronary heart disease based on Co-expression Network

Authors:

dongxue Wu (18810820682@163.com)
Mengqi Huo (20150931715@bucm.edu.cn)
Xi Chen (chenxi_cx95@163.com)
Yanling Zhang (zhangyanling@bucm.edu.cn)
Yanjiang Qiao (yjqiao@bucm.edu.cn)

Version: 2 Date: 05 Oct 2019

Author’s response to reviews:

Dear Editor:

Thank you very much for considering the revised version of our manuscript entitled "Mechanism of tanshinones and phenolic acids from Danshen in the Treatment of Coronary heart disease based on Co-expression Network" (BCAM-D-19-00357R1) to BMC Complementary and Alternative Medicine for publication. We are grateful to the reviewers and the Editor for pointing out some important modifications needed in the report. The explanation of what we have changed in response to the essential revisions is given point by point below.

1. We note you have not cited Figure 5 within your manuscript. Any figures, table or additional files which are not cited within your manuscript may be removed our production team.

Response: The citation of Figure 5 has been added in the manuscript.

2. In order to be in line with journal requirements, please include the following headings within your manuscript: Background, Methods, Results, Discussion, Conclusions

Response: The the following headings: Background, Methods, Results, Discussion, Conclusions were added in the manuscript.

3. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Response: The description about the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript was added in the Funding section.

4. Please put your responses to the reviewers' editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: Thanks for your reminding. We have put the responses to the editors’ comments in the Response to Reviewers box in Editorial Manager. The manuscript was uploaded as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files were clean versions. Additional files were uploaded as separate files. All figures, tables and additional/supplementary files are cited within the text.